Author's response to reviews

Title: Widening access to medicine may improve general practitioner recruitment in deprived and rural communities: survey of GP origins and current place of work

Authors:

Jon Dowell (j.s.dowell@dundee.ac.uk)
Michael Norbury (michaelnorbury1@aol.com)
Kathryn Steven (kathryn@doctors.org.uk)
Bruce Guthrie (b.guthrie@dundee.ac.uk)

Version: 6  Date: 11 September 2015

Author's response to reviews: see over
27.07.2015

Dr Malcolm Boyle,
Editorial Board,
BMC Medical Education

Dear Dr Boyle

**BMC Medical Education MS: 1629096781645959 – Widening access to medicine may improve GP recruitment in deprived and rural areas**

Thank you for asking us to review the authors’ contributions for this paper as directed by the handling editor. The ‘Author contribution’ section has been revised as can be seen on page 10, lines 231-239.

“JD conceived the research study, defined the research question and methodology. JD was also involved in writing of the paper and approved the final manuscript.

MN conceived the research study, defined the research question and methodology and was involved in data collections and analysis. MN was also involved in writing of the paper and approved the final manuscript.

BG conceived the research study, defined the research question and methodology and was involved in data collections and analysis. BG was also involved in writing of the paper and approved the final manuscript.

KS was involved in writing the paper and revised the manuscript critically prior to preparing the final paper for submission.”

Many thanks for your ongoing assistance with this paper.

Yours sincerely,

Jon Dowell on behalf of all the authors